Author’s response to reviews

Title: Clinical significance of HER2 and EGFR expression in Colorectal Cancer Patients with Ovarian Metastasis

Authors:

Jilin Li (li_jilin@foxmail.com)
Shu-Han Lin (linshuhan52@vip.sina.com)
Hong-Qiu Chen (chenhongqiu2012@163.com)
Li-Sheng Liang (Lianglisheng2007@163.com)
Xian-Wei Mo (mo_xianwei@126.com)
Hao Lai (laihao066@sina.cn)
Jie Zhang (18777194753@163.com)
Jing Xu (1045608666@qq.com)
Bing-Qian Gao (cityhunter27@sina.com)
Yan Feng (emily_feng@foxmail.com)
Yuan Lin (doctoryuanlin@sina.cn)

Version: 2 Date: 17 Feb 2019

Author’s response to reviews:

Dear Editors:

Thank you very much for your response.

We have revised our manuscript according to your comments. A list of abbreviations was added after the Conclusions section. The headings and the authors’ contribution section were revised. A statement indicating all authors read and approved the final version of the manuscript was added as required.
All authors have read the final version of submission and a clean version of revised manuscript was submitted.

We deeply appreciated your consideration of our manuscript again.

Yours Sincerely,

Yan Feng

Point to point response:

1. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

   The list of abbreviations was added after the Conclusions section.

2. Please format your Manuscript to include the following headings: Background, Methods, Results, Discussion and Conclusions.

   Revised according to your comments.
3. Please note that all authors should be named by using their initials rather than full names, and also include a statement clearly indicating that all authors read and approved the final version of the manuscript, in the Authors' Contributions section.

The authors' names were revised and the statement “All authors read and approved the final version of the manuscript.” was added at the end of the authors’ contributions section.

4. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

The clean version was submitted.